Author's response to reviews

Title: Multimorbidity Patterns and Health Service Use in Swedish 85 year olds: An Exploratory Study

Authors:

Huan-Ji Dong (huanji.dong@liu.se)
Ewa Wressle (ewa.wressle@lio.se)
Jan Marcusson (jan.marcusson@liu.se)

Version: 2 Date: 29 May 2013

Author's response to reviews:

A summary providing a point-by-point response to each comment is enclosed in the cover letter.